JGR Atmospheres

RESEARCH ARTICLE

10.1029/2022JD038473

Key Points:

- diurnal warm layers (DWLs) increase atmospheric moisture
- The increase of cloud cover (CC) following the formation of a DWL is immediate and only lasts for several hours
- The magnitude of the CC increase is small and has no discernible influence on the global mean

Correspondence to:

R. Shevchenko, radomyra.shevchenko@mpimet.mpg.de

Citation:

Shevchenko, R., Hohenegger, C., & Schmitt, M. (2023). Impact of diurnal warm layers on atmospheric convection. *Journal of Geophysical Research: Atmospheres, 128*, e2022JD038473. https://doi.org/10.1029/2022JD038473

Received 31 DEC 2022 Accepted 26 JUN 2023

© 2023. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Impact of Diurnal Warm Layers on Atmospheric Convection

Radomyra Shevchenko¹ , Cathy Hohenegger¹, and Mira Schmitt²

¹Max Planck Institute for Meteorology, Hamburg, Germany, ²Leibniz Institute for Baltic Sea Research, Warnemünde, Germany

Abstract This manuscript presents a study of oceanic diurnal warm layers (DWLs) in kilometer-scale global coupled simulations and their impact on atmospheric convection in the tropics. With the implementation of thin vertical levels in the ocean, DWLs are directly resolved, and sea surface temperature (SST) fluctuations of up to several Kelvin appear in regions with low wind and high solar radiation. The increase of SST during the day causes an abrupt afternoon increase of atmospheric moisture due to enhanced latent heat flux (LHF), followed by an increase in cloud cover (CC) and cloud liquid water (CLW). However, although the diurnal SST amplitude is even exaggerated in comparison to reanalysis, this effect only lasts for 5–6 hr and leads to an absolute difference of 1% for CC and 0.01 kg m⁻² for CLW. This can be explained by the fact that the low wind over the SST anomalies dampens their potential effect on the LHF and hence clouds. All in all, the impact of DWLs on convective CC is found to be negligible in the tropical mean.

Plain Language Summary The diurnal fluctuations of sea surface temperature (SST) have been extensively studied for the last decades, but the assessment of the importance of this phenomenon for atmospheric convection on the global scale has come within reach only very recently, thanks to the development of simulations with a horizontal resolution of O(1 km). In this manuscript we show that we can indeed observe an impact of SST fluctuations on moisture in the atmosphere. However, the impact on the amount of clouds in the tropics is found to be short-lived and its magnitude negligible on average.

1. Introduction

Diurnal sea surface temperature (SST) anomalies and their interplay with the atmosphere and in particular with the diurnal cycle of convection have been an object of study for many decades. In this study, we investigate this connection for the first time using simulations that can explicitly resolve both the diurnal temperature variations in the ocean and convection in the atmosphere on a global scale.

Diurnal variations in SST have already been described in Sverdrup et al. (1942). Since then, there have been numerous studies describing the physics and the conditions of appearance of diurnal SST variations, the seminal work by Price et al. (1986) being the first detailed description of this phenomenon. Under low-wind conditions and with sufficient insolation, a stable near-surface layer forms during the day in the upper layers of the ocean (until the depth of O(10 m)) that leads to a surface warming of up to 5 K (see Wick and Castro (2020)). In the absence of solar radiation during the night, the stratification dissolves as vertical turbulent mixing takes overhand, until a homogeneous mixed layer is restored. The physics of this phenomenon is described in detail in a monograph by Soloviev and Lukas (2013). This stratified, warm layer is known as diurnal warm layer (DWL) and it is ubiquitous in all latitudes, causing SST fluctuations of 0.2 K or more in the entire Northern hemisphere and beyond during boreal summer (see Gentemann et al. (2003)). A comprehensive discussion of its definition and properties can be found in a review by Kawai and Wada (2007). In particular, the authors of the review point out that the presence of DWLs in observations as well as in single column simulations leads to stronger latent and sensible heat fluxes. As surface fluxes connect the surface to the atmospheric boundary layer and since changes in boundary layer properties affect the development of convection, the question of the impacts of DWLs on atmospheric convection arises.

Investigating this question in models requires both fine enough vertical resolution in the ocean, to resolve DWLs, and fine enough horizontal grid spacing in the atmosphere, to resolve atmospheric convection. With the development of deca- to kilometer scale simulations in a coupled configuration (Hohenegger et al. (2023)) such investigations are becoming possible. Prominent among the newest studies are the papers by Voldoire et al. (2022) and Brilouet et al. (2021). In Voldoire et al. (2022), a single column coupled model has been considered, while





in Brilouet et al. (2021), a one column ocean model has been coupled to an atmospheric large-eddy simulation model integrated over a limited area. Both experiments are based on or validated with the data from the Dynamics of the Madden Julian Oscillation (DYNAMO) campaign, during which diurnal SST differences of several Kelvin were observed. Voldoire et al. (2022) showed that the impact of DWLs on the boundary layer depth, atmospheric moisture and precipitation seems to be small. In contrast, de Szoeke et al. (2021) argued that in the observations from the DYNAMO data set, convection is enhanced on days with large SST differences. Finally, Voldoire et al. (2022) conjectured that a single column model cannot capture horizontal interactions that might lead to a larger impact.

To the authors' knowledge, there is yet no study extending the question of interactions between DWLs and atmospheric convection to a realistic, global framework with resolved convection. And yet, atmospheric convection plays a key role in the energy and water cycle in the tropics, therefore describing the driving mechanisms of convection and assessing their importance is crucial.

The present study aims at closing this gap and at analyzing the impact of DWLs on atmospheric convection in a global, coupled ICON simulation with 5 km horizontal resolution and decimeter grid spacing in the first oceanic layers. This allows us to resolve DWL and convection, to assess their interactions, also by resolving horizontal gradients, and to analyze the global impacts of DWLs. We are particularly interested in understanding whether the presence of DWLs enhances CC in a time frame of several days and if so, whether this happens through a direct moistening by the latent heat flux (LHF) or by enhancing spatial gradient in SST and mesoscale circulations. Moreover, the global nature of our simulation allows to zoom in on different areas and encompass other meteorological and sea conditions than the ones of the DYNAMO campaign.

The manuscript is structured as follows. In Section 2 we describe the setup of the experiments, and in Section 3 we analyze the representation and properties of DWLs in global ICON simulations. In Section 4 we quantify the effect of DWLs on surface fluxes, atmospheric moisture and clouds, and finally in Section 5 we present the conclusions of our work.

2. Experimental Setup

To study the effect of DWLs, we conduct global coupled simulations with the ICON model in its Sapphire configuration. The Sapphire configuration targets simulations with a horizontal grid spacing finer than 10 km. For our simulations, we use a setup similar to the simulation called G_AO_5 km in Hohenegger et al. (2023), with small deviations that will be described below. The model is fully coupled, and at the horizontal resolution of approximately 5 km it is at the boundary of resolving convection. Although the cloud amount associated with shallow convection is expected to be overestimated (see Vial et al. (2019) and Hohenegger et al. (2020)), it has been demonstrated in Vial et al. (2019) that both amplitude and shape of the diurnal cloud cycle remain similar for horizontal resolutions varying between 150 m and 2.5 km in ICON simulations. This setup allows us to directly assess how an SST anomaly influences convection, to study possible interactions between clouds and DWLs in a highly realistic context, and to analyze global implications of including DWLs into models.

The main prerequisite for resolving DWLs in ocean simulations is high vertical resolution of the upper oceanic layers (cf. Brilouet et al. (2021)). The introduction of the z^* ocean coordinate into the ICON model (detailed in Singh and Korn (in preparation)) allows running global experiments with an unprecedented vertical resolution. For the purposes of this article, we conducted two simulations. One has 128 ocean vertical levels, starting at 2 m at the surface and gradually growing with depth, which is also the setup in G_AO_5 km in Hohenegger et al. (2023). This simulation plays the role of the reference simulation. The other has 139 ocean vertical levels, starting with 0.5 m and gradually increasing and blending into the reference setup from 45.4 m depth (see Figure A1 in Appendix A). To determine layer thicknesses necessary for resolving DWLs, while keeping the total number of layers as low as possible, we ran test simulations with the one dimensional ocean model GOTM (Umlauf et al. (2005)). The vertical grid that we chose is sufficiently coarse to obtain a numerically stable global run and produces DWLs whose temperature field is indistinguishable in its depth, magnitude and overall vertical structure from a run with a 10 cm uniform grid (compare Figure A2 in Appendix A and Figure 1). We run both simulations for 30 days starting 21 January 2020 and with the coupling and radiation time steps of 12 min, the atmosphere time step of 30 s, and the ocean time step of 80 s. Both simulations have identical initial conditions, whereby the





Figure 1. 48 hr simulation of diurnal warm layers in GOTM with 139 ocean vertical levels (see Figure A1 for the detailed vertical resolution), as will be used in ICON, for an idealized heat flux with a maximal radiation of 600 W m⁻², a constant wind speed of 8 m s⁻¹, and a time step of 80 s.

generation of the initial ocean state is described in Hohenegger et al. (2023). We call the runs with 128 and 139 ocean levels $S_{control}$ and S_{+DWL} respectively.

3. Representation of Diurnal Warm Layers

In this section we describe the features of DWLs as represented in our simulations and compare them to known properties derived from measurements, reanalysis, and limited-area decameter-scale simulations. As our main focus is on tropical convection, we concentrate our analysis on the tropics which we define here as the area comprised between 40° N and 40° S.

3.1. Occurrence and Magnitude

First we diagnose DWLs and assess their occurrence and magnitude. Rigorously speaking, DWLs are defined via vertical temperature gradients (see e.g., Matthews et al. (2014)), as they represent a temperature anomaly sitting on top of the ocean mixed layer. Instead of the temperature gradient, the

diurnal SST amplitude (DSA) is often used as a proxy (e.g., in Voldoire et al. (2022)), because, as mentioned in Bellenger and Duvel (2009), in the tropics one can assume that horizontal advection happens at longer time scales, and the temperature changes in the upper ocean can be attributed to DWLs. We also adopt this approach, formally defining DWLs as places with DSA larger than 0.1 K, which corresponds to non-zero depth of a DWL in Matthews et al. (2014). Care is required when following this approach, as SST in our model is defined as the temperature in the uppermost layer. In S_{+DWL} it is the average temperature over 0.5 m and in S_{control} the average temperature over 2 m. Since in this study our main interest is the response of the atmosphere, and since this is the temperature that the atmosphere sees, we do not correct for this difference in our analysis.

In our simulations, DWLs are ubiquitous: Even spots with DSA of over 1.5 K cover 5% of the tropical ocean area. Comparing DWLs in our simulations and in ERA5 reanalysis, we can see in Figure 2 that for the last 10 days of January the spatial distribution of DWLs is in moderately good agreement with each other, with a correlation coefficient of 0.51. The areas with large DSA are well represented by both simulations, but in the less active areas DWLs seem to appear more often in our simulations than in ERA5, creating a higher average. The major hot spots of DWLs are in the Indian ocean south of the equator, along the western coasts of America and Africa, and in the southern part of the Pacific, both in our simulations and in ERA5. Similar areas are also identified in the January climatology map of Bellenger and Duvel (2009). The discrepancy between the DSA pattern in simulations and ERA5 might be due to the divergence in atmospheric conditions over 10 days as well as to local biases in near-surface winds (not shown). A striking feature of our simulation is that, while for S_{control} the DSA field appears nearly homogeneous and the extremes are underestimated compared to ERA5, as expected from the use of thick ocean vertical layers, the amplitudes in S_{+DWL} are much larger than in the reanalysis, with values twice as high. Our values are in contrast with observations as summarized in Kawai and Wada (2007), where the average DSA values during the boreal winter rarely surpass 1 K. Since Kawai and Wada (2007) employed the skin temperature to diagnose DWL, we can conclude that DSA is overestimated in S_{+DWL} . This is confirmed by further observations. The PIRATA buoys located at 0°N10°W and 0°N0°E at 1 m depth have each measured DSA of under 0.7 K in the same period, and the average DSA measured during the EUREC⁴A field campaign in the area $56.5^{\circ}W-59^{\circ}W$ and $12^{\circ}N-14.5^{\circ}N$ is about 0.15 K, while in S_{+DWL} the values at these locations are 1.8 and 0.71 K respectively. A possible cause for the overestimation is insufficient vertical mixing in the upper layers of the ocean. As this study concentrates on the atmospheric effect of the DWLs, this problem does not jeopardize the analysis, and if anything indicates that the simulated effects will be too strong.

Another important feature is that in S_{+DWL} , DWLs tend to avoid areas with high CC: on a given day, 24% of cells over the ocean have a CC of 0.9 or higher, while for cells where DWLs develop this number reduces to 13%. However, there are also many areas with a low CC and yet no significant DWLs, as shown in Figure 3. For instance, out of all areas with CC below 0.3 on a given day, only 9% developed DWLs with DSA of 1.5 K or higher. This is a consequence of a property well documented in observational studies. Indeed, as explained in Soloviev and Lukas (2013), shortwave radiation and surface wind are the two principal driving factors of DWLs, where high shortwave radiation and low wind speed favor the appearance of DWLs. The previously mentioned





Figure 2. Magnitude of the diurnal SST amplitude (max-min), averaged between 22 and 30 January 2020 in (a) $S_{control}$, (b) S_{+DWL} , and (c) ERA5. The white rectangle in (b) designates an area in the Indian ocean that is analyzed in Section 3.1, and the black rectangle designates the EUREC⁴A campaign area.

areas with low CC but no DWL correspond to high-wind zones. The relationship of DWLs with downward shortwave radiation and near-surface wind speed will be further explored in Section 3.2.

As to the horizontal extent of DWLs as simulated by S_{+DWL} , there are two observations to be made. As shown in Figure 4a, there are many small clusters of DWLs, but the total area they cover is practically negligible: although 177 clusters out of 799 are of size $\leq 625 \text{ km}^2$, the total area covered by them amounts to only 0.3% of the entire area covered by DWLs. On the opposite end of the histogram in Figure 4a, one can see that there are a few clusters of size 10^6 km^2 . They form predominantly in the high DSA areas from Figure 2 (not shown).









Figure 4. Diurnal warm layer (DWL) statistics for S_{+DWL} : (a) clusters of DWLs to the threshold of 1.5 K on a particular day (23 January 2020) ordered in a histogram of cluster sizes, (b) normalized histogram of DWL episode lengths to different thresholds for the box 70°E-75°E and 5°S-10°S (see white box in Figure 2) over a period of 30 days.

Moreover, for each particular grid cell, the DWLs do not seem to be persistent: for instance, in a region with very high DSA in the Indian ocean (between $70^{\circ}E-75^{\circ}E$ and $5^{\circ}S-10^{\circ}S$, see the white square in Figure 2b) during a period of 30 days no episode of DSA >0.6 K lasted longer than 10 days, and 80% of all episodes lasted 5 days or less (see Figure 4b). Expectedly, the persistence time becomes shorter if a higher threshold for DSA is considered. This result is similar to the findings in the ERA40 forced simulation in Bellenger and Duvel (2009), where they showed that the DWL episode duration decays faster than exponentially. The lack of persistence applies even to large clusters that can completely change their location from 1 day to the next (not shown).

3.2. Structure of DWLs and Preconditions for Their Formation

For a deeper understanding of how the known properties of DWLs are captured by S_{+DWL} we focus on a study of this phenomenon in the tropical northern Atlantic region off the coast of Barbados, more precisely in the box 56.5°W-59°W and 12°N-14.5°N corresponding to the region of the EUREC⁴A field campaign (Stevens et al. (2021)).

First, we examine the vertical structure of a DWL produced in the simulation S_{+DWL} . As shown in the anomaly profile (Figure 5), the incoming heat quickly accumulates in the upper layers during the day, as the warming of the upper layers creates a stable stratification if the mixing is slow, which is the case under low wind condi-



Figure 5. Sea water temperature (3 hr mean output) averaged over 3 days over the box $56.5^{\circ}W-59^{\circ}W$ and $12^{\circ}N-14.5^{\circ}N$, time anomaly.

tions. During the night, the DWL slowly dissipates until a single mixed layer is restored. As mentioned before, in this region, the heat anomalies are overestimated: a typical DSA in glider observations is approximately 0.15 K (see Hohenegger et al. (2023), Figure 11c), while in S_{+DWL} in a calm period it is around 0.3 K. The time of the maximum and minimum is captured accurately compared to these observations (between 14 and 17 hr LT and between 5 and 8 hr LT respectively).

Another analysis demonstrates that the dependency of DSA on insolation and surface wind observed, among others, in satellite measurements in Gentemann et al. (2003), are reproduced in ICON simulations as expected, namely higher DSA correlates with lower winds and higher downward shortwave radiation at the surface. The scatterplot in Figure 6a shows a relationship similar to the formula

DSA ~
$$\begin{cases} U^{-3} \text{ for } U < 4\text{-}5\text{m s}^{-1}, \\ U^{-1} \text{ for } U > 4\text{-}5\text{m s}^{-1}, \end{cases}$$





Figure 6. 2D histograms of sea surface temperature amplitude and (a) 10 m wind speed as well as (b) shortwave radiation over the box 56.5°W to 59°W and 12°N to 14.5°N in S_{+DWL} . The blue and red lines in (a) indicate the interpolation of U^{-1} and U^{-3} to higher and lower wind speeds respectively. The red line in (b) is the linear regression fit between the variables.

suggested in Soloviev and Lukas (2013) for 10 m wind speed U, although the threshold appears to be located at $6-7 \text{ m s}^{-1}$ rather than at $4-5 \text{ m s}^{-1}$. Also the linear dependence on shortwave radiation described in Matthews et al. (2014) seems to be confirmed in S_{+DWL} (Figure 6b). In particular, the slope of the linear regression between DSA and surface downward shortwave radiation in ICON is 0.0023 K W⁻¹ m², which is indeed close to the slope of 0.0021 K W⁻¹ m² found in Matthews et al. (2014). The large scattering of points in Figure 6b is caused by the joint influence of wind speed and radiation on DWLs.

4. Effect of DWLs on Convection

The previous section shows that DWLs are frequent in the tropics, they cover large areas and persist over several days. We can therefore hypothesize that they might have an effect on atmospheric convection, either by creating horizontal air temperature gradients, or by enhancing the moistening of the overlying air. Moreover, as DSA in S_{+DWL} is overestimated, the effect of DWLs is expected to be too strong.

To assess the impact of DWLs on convective clouds, we need to compare the values of relevant variables in places where DWLs develop to places where they do not develop. A clean way to accomplish this without having to account for other sources of variability is to compare the differences between these values in S_{+DWL} and $S_{control}$ at places where DWLs of a certain magnitude are present in S_{+DWL} . A key point in this analysis is to only include the first 4 days after the start of the simulation into the investigation to ensure that the two simulations remain close and do not diverge due to the chaotic nature of the atmosphere. Figure 7 shows that both simulations indeed remain highly correlated during the first several days, and this correlation is well above a simple measure of persistence, computed here as 1-day lag correlation in $S_{control}$.

To conduct our analysis, we proceed as follows. As we are interested in the temporal evolution of the effects of DWLs, we move the entire output to local time. The analysis then starts at a given point on the first full day, making sure that at every grid cell a spin up period of at least 12 hr can be discarded. The ocean is then subdivided into $0.25^{\circ} \times 0.25^{\circ}$ disjoint squares, and for each square the hourly maximal SST difference between S_{+DWL} and $S_{control}$ during a day, denoted by Δ SST, is computed. Based on this first day of the analysis that we also call the detection day, two groups can be formed: one called the "no DWL difference" group, where Δ SST stays below 0.1 K, and another one where Δ SST lies between certain thresholds T_1 and T_2 , which we call the "DWL difference between T_1 and T_2 " group. Note that the plot of Δ SST yields a map very similar to Figure 2b (Figure B1 in Appendix B), and thus can be used as a proxy for identifying DWL areas. We use this proxy from now on to cover a larger variety of cases and to obtain a cleaner comparison.



Journal of Geophysical Research: Atmospheres



Figure 7. Temporal evolution of the global correlation between $S_{control}$ and S_{+DWL} (black line) and of $S_{control}$ with itself shifted by 24 hr (green line) for (a) near-surface wind, (b) cloud cover, and (c) incoming shortwave radiation at the surface. The vertical line at 6 hr LT designates the approximate beginning of the formation of diurnal warm layers.

The subsequent analysis consists in comparing the two groups to each other, focusing on the time series of the differences in relevant variables over the days 1–3, with day 1 being the detection day, see Figure 8 for an example of the differences in SST. Statistically, it is supported by *t*-tests for equality of means for independent samples with different variances. We consider groups with T_1 and T_2 increasing in steps of 0.1 K, every $T_2 - T_1 = 0.1$ K group having at least 1,695 cells, a sufficient number to ensure the robustness of the results. The size of the squares is chosen in accordance to the cluster sizes discussed in Section 3.1 (see Figure 4), as a mesoscale analysis is sufficient to capture a possible effect of DWLs, and also because the temperature and moisture changes produced by one isolated output cell are not expected to have a large effect and would be quickly mixed in the atmosphere. Moreover, we make the assumption that the effects in cloudiness are localized. Indeed, the impact of DWLs on CC via moistening is a local effect, and the mesoscale circulation created by SST gradients would also result in cloud formation over DWLs. With this assumption, the samples can be treated as independent. We analyze the results over 3 days, because we want to study the overall duration of the atmospheric reaction in the areas where there were DWLs and catch possible reinforcement phenomena under realistic persistence conditions of DWLs. To avoid capturing temporally correlated events, we also refrain from treating the output shifted by a day or more as additional samples.



Figure 8. Time series of S_{+DWL} - $S_{control}$ for different ΔSST .

4.1. Impact on Latent Heat Flux and Related Quantities

First we investigate whether higher Δ SST leads to more moisture in the air over DWL areas. This will help understand whether cloud formation may be impacted over DWLs. In particular, we are going to concentrate on differences in LHF and column water vapor path (WVP) to measure the effects on evaporation and column moistening respectively, as shown in Figure 9.

The first observation concerning the plots (a, b) in Figure 9 is that the differences between S_{+DWL} and $S_{control}$ in the "no DWL difference" group after the detection day revolve around zero, as expected and further justifying our approach. This is not the case for the "DWL" groups. Differences are clearly appearing and are of larger magnitude, the stronger the DWL is.

The LHF is increased when DWLs are present, with a peak around noon on every day, as expected from the diurnal cycle in SST. WVP is always higher over DWLs. The difference decreases in the morning hours of the detection day and increases rapidly thereafter. The dip could be caused either by a dip in LHF, by an increase of precipitation or by a weakening in convergence of moisture. As LHF is increased and there is no excessive rain over DWLs (not shown), the dip must be due to weaker convergence, which might be caused by a weakening of surface wind required for DWL formation. The increase of





Figure 9. Time series of S_{+DWL} - $S_{control}$ and *t*-test significance levels in bins of 0.1 K for latent heat flux (a, c) and water vapor path (b, d). On panel c and d, white means not significant.

WVP after 12 hr LT reflects the increase in moisture due to increased LHF. One can trace the propagation of the moisture response to DWLs in Figure 9, since the onset and duration times of the effects are different: while LHF reacts immediately and is tied directly to differences in temperature (see Figure 8), the effect on WVP is delayed by several hours and persists over the following days without losing its magnitude.

Figures 9c and 9d confirms that the differences observed visually are highly significant even on the third day and even for small Δ SST with p-values smaller than 0.005, except for the nighttime values of LHF. However, an important point is that the absolute differences are rather small, especially for smaller Δ SST: For instance, for Δ SST between 0.6 and 1.0 K, the maximum difference of LHF is below 15 W m⁻² and WVP is below 0.2 kg m⁻². One can compare these values with global averages of the two quantities over the ocean, which are ca. 130 W m⁻² for LHF and ca. 34.6 kg m⁻² for WVP. The order of magnitude of LHF changes is similar to that found in the single column configuration in Voldoire et al. (2022).

In Acquistapace et al. (2022), the authors described an observational case study in the EUREC⁴A region that compares the atmospheric response to a mesoscale cold patch with 1 K difference to other parts of the ocean in this region. Although this difference in SST is comparable to the differences caused by DWLs, the impact on the LHF in this study is much higher than the values shown in Figure 9a and has on average 70 W m⁻² higher values in the subcloud layer over the warmer area. The difference between the two studies could come from the different height use to compute the LHF, at the surface versus in the subcloud in Acquistapace et al. (2022). This is nevertheless unlikely as LHF only slightly increases with height (see Siebesma et al. (2003)). The other explanation could be the wind. The principal difference between DWLs and an SST anomaly of a different origin is that the appearance of DWLs requires windstill conditions, and this has an influence on LHF, as it scales linearly with wind. Indeed, the winds in Acquistapace et al. (2022) are about 10 m s⁻¹, while in our case, winds in



the 1 K difference group are on average 2.75 m s⁻¹. Given that in the 1 K difference group the LHF difference is 18 W m⁻² (see Figures 8 and 9), a three times higher wind would have given values similar to Acquistapace et al. (2022). We can conclude from this simplified calculation that the wind is an important inhibition mechanism for the possible effect of SST anomalies caused by DWLs on the atmosphere.

4.2. Impact on Cloudiness

The variables we consider for analyzing the impact of DWLs on clouds are total CC and cloud liquid water (CLW) path (CLW). In our configuration of the ICON model, the CC values are binary, so the values over 0.25° boxes represent the fraction of the gridpoints in that box covered by clouds.

The responses of CC and CLW shown in Figure 10 are very similar to each other. The "no DWL difference" time series is fluctuating around zero after the detection day, but on the detection day itself there is a peak at midday. In contrast, the DWLs regions exhibit dips at the same time, and the magnitude of these dips is positively correlated with the Δ SST amplitude (see Figures 10a and 10b and Figures 10e and 10f). This illustrates the prerequisite for DWLs to exist mentioned in Section 3.1: The appearance of DWLs is generally favored by a lower cloud amount. Subsequently we can observe the cloud response to the formation of DWLs on the detection day: after about 12 hr LT, the difference between S_{+DWL} and S_{control} in both CC and CLW over DWLs starts to grow, and by 15 hr LT the peak of discrepancy is reached. This growth is statistically significant starting from an SST amplitude of 0.3 K, and the effect persists longer for higher values of Δ SST, namely until 20 hr LT for Δ SST of 1.5–2.0 K and until 18 hr LT for Δ SST of 0.3–1.0 K for CC. Moreover, the effect seems to last longer for CLW, although the values of both variables fall back below the zero line already during the night. In total, the increase of cloud following the formation of a DWL lasts up to 5–6 hr.

A behavior similar to the first day, that is, less CC and CLW in S_{+DWL} compared to $S_{control}$ in the morning and rise of these quantities in S_{+DWL} in the afternoon, can be observed also on the second and third day after the detection day (Figures 10a and 10b), however, the deviation is barely significant: most of the time, the p-value is above 0.05, especially for higher values of Δ SST (Figures 10c and 10d). This means that the increase of the quantities is often not homogeneous enough to be interpreted as systematic (see Figures 10c and 10d). A crucial factor here is the absolute magnitude of the differences: In extreme cases, for Δ SST between 1.5 and 2.0 K, the deviation of CC reaches 0.03 and that of CLW 0.02 kg m⁻² on the detection day. On the following day the corresponding maximal values fall to 0.01 and 0.01 kg m⁻², while the averages are 0.001 and 0.002 kg m⁻² respectively (see Figures 10a and 10b). We can compare those to the average values of CC and CLW over DWLs, 0.65 for CC and 0.11 kg m⁻² for CLW. The average increase in CLW is about 1% and the increase in CC is about 0.2% on the day following even very large Δ SST. We can therefore conclude that the effect on the cloud water content is higher than the contribution of DWLs to CC change, but the effects are very small.

The morning/midday dip of CC and CLW over DWLs even overcompensates the subsequent rise of these quantities. Figure 11 shows mean CC differences between S_{+DWL} and $S_{control}$ as a function of Δ SST on the detection day. For Δ SST above 0.1 K the simulation S_{+DWL} has lower values of CC than $S_{control}$, and only for Δ SST below 0.1 K the reverse is true, such that on average, S_{+DWL} has slightly less cloud than $S_{control}$. The effect of DWLs appears to be dominated by the variability of the model. All in all, we can conclude that DWLs do not increase the global mean of CC or CLW path.

We continue the analysis by looking at deep and shallower clouds separately, as one might expect to see more effects associated with one type of clouds. We differentiate between deep and shallow clouds by taking the outgoing longwave radiation at the top of the atmosphere of 240 W m⁻² as a threshold. This value is mentioned in Fu et al. (1990) as the threshold often used to identify deep convection. The figures for shallower clouds are very similar to Figure 10 (not shown), which can be explained by the fact that most DWLs are found outside of deep convective regions, see Figure 2b or Figure B1. The deviation for CLW in deep clouds is somewhat larger, up to 0.03 kg m⁻² on the detection day. As shown in Figure 12a for CC, the effects, despite being a bit larger, remain very small and do not exhibit systematic significance (Figure 12c). Considering precipitation as an additional possibility to see an impact of DWLs (Figures 12b and 12d), we see that there is no sufficient evidence to attribute the fluctuations of precipitation, even for higher values of Δ SST, to anything more than chance.

The global results obtained in this section for the CC, CLW, and precipitation seem robust with respect to the chosen large-scale area. In particular, excluding the subtropics has no effect on the amplitudes or the temporal





Figure 10. Time series of S_{+DWL} - $S_{control}$, *t*-test significance levels in bins of 0.1 K, and maximal (red) and minimal (blue) value of the deviation on the detection day for cloud cover (a, c, e) and cloud liquid water (b, d, f).

behavior of these quantities. Moreover, the results remain quantitatively unchanged if the simulations are compared after additional 24 hr of spinup (not shown).

The last question that we want to investigate is to what extent the presence of DWLs might affect the diurnal cycle of convection. We focus on the example of the EUREC⁴A field campaign, which took place in the northern tropical Atlantic. For this region the diurnal cycle of shallow convection has been studied in great detail in Vial et al. (2019). In particular, it has been shown there that, in observations as well as in large-eddy simulations (LES), the CC peaks during the day, which might make it more susceptible to SST in comparison to other regions. In our case, both S_{+DWL} and $S_{control}$ overestimate the CC found in observations by a factor of 2, as was expected from the resolution sensitivity study in Vial et al. (2019), but capture perfectly the time structure as well as the amplitude of the cycle: minimum between 12 and 15 LT, maximum at 2 LT, amplitude of about 0.1 (Figure 13), in good agreement with observations in Vial et al. (2019). The increase in CC resulting from the appearance





Figure 11. Diurnal mean cloud cover (CC) differences $(S_{+DWL}-S_{control})$ as a function of Δ SST (black crosses). The red cross is the average CC difference against the average Δ SST.

of DWLs occurs in the late-afternoon. If anything, CC slightly decreases at other times. Both effects taken together tend to slightly reduce the diurnal amplitude. The fact that DWLs only have a small influence on the diurnal cycle of CC is consistent with the study in Vial et al. (2019) that reproduced the main features of the cycle in LES despite using an SST with no diurnal cycle.

5. Discussion and Conclusions

By introducing thin vertical levels into the global coupled ICON model, we could directly resolve DWLs and assess their impact on the atmosphere. The simulations employed a grid spacing of 5 km, both in the atmosphere and ocean, so that ocean mesoscale eddies and atmospheric convection can be resolved explicitly. The DWLs produced by the simulation reproduce the physical features known from observations and limited area decameter simulations, but the magnitude of the diurnal SST fluctuations is exaggerated in comparison to reanalysis, by about a factor of two.

The increase in the amplitude of the diurnal cycle of SST in regions with DWLs leads to a corresponding increase in LHF and WVP. The effects are significant, even on days 2 and 3 following the detection of a DWL, but the values are small: 7 W m⁻² difference in LHF and 0.1 kg m⁻² difference in WVP for a SST difference of 0.6 K. In the late-afternoon of the detection day, CC and CLW content also increase, but the effects are small and lose statistical significance within 5–6 hr of appearance. What is however significant is a decrease in CC and liquid water content on the day of detection, around noon. This expresses the fact that DWLs favorably form in areas of



Figure 12. As Figure 9, but for cloud cover (a, c) and precipitation (b, d) associated with deep convection.





Figure 13. Diurnal cloud cover cycle in the box 56.5°W to 59°W and 12°N to 14.5°N, composite over 30 days.

low CC, and hence high insolation. This effect compensates the subsequent increase. All in all, resolving DWLs does not affect the mean CC over tropical oceans.

The amplitude of the observed differences in LHF is similar to findings in Voldoire et al. (2022), and the impact on CC shows that convection over DWLs is enhanced, as it is claimed in the observation study of de Szoeke et al. (2021). Moreover, the small influence of DWLs on the CC diurnal cycle in the tropical Atlantic supports the results in Vial et al. (2019). A surprising finding of our study is that the impact of DWLs on convection remains small even for a strongly enhanced diurnal SST amplitude over this particular region as well as globally. A possible reason for this is that DWLs develop in windstill areas, and their impact on the LHF therefore remains small, in contrast to the observational study in Acquistapace et al. (2022) showing a large atmospheric effect of a 1 K SST anomaly under high wind (10 m s⁻¹). In view of this crucial difference, we conclude that it is important to resolve (sub)mesoscale oceanic structures such as eddies that result in large SST anomalies even under strong wind conditions.

We finish the discussion by focusing on the implications and limitations of our study. In the subtropics, the resolution of 5 km can strongly distort the representation of shallow convection. Although the similar behavior between the different resolution simulations in Vial et al. (2019) is encouraging, we cannot exclude at the outset that our results may be biased by a poor representation of shallow convection. Still, we could also not find any significant effect of DWL on deep convection, which suffers less from the limitation in grid spacing.

Regarding the question of the importance of DWLs in models, one needs to differentiate between the local and the overall impact. In our exemplary study in the northern tropical Atlantic, the presence of DWLs can reduce the amplitude of the diurnal cycle of the CC by up to 10% (see Figure 13). In some specific cases with extraordinarily high DSA, DWLs might indeed play a role. However, this remains a rare and local phenomenon.

The analysis in this manuscript only concerns short-term effects of DWLs. However, it is known that the mean SST increases in DWL areas (Bellenger and Duvel (2009)). Therefore, inclusion of DWLs will have a long-term influence on the energy budget that is not treated here, but at least for short-term effects, our study demonstrates that DWLs do not have a global and significant impact.

Appendix A: Vertical Resolution of the Simulations

Figure A1 demonstrates the difference in vertical resolutions between S_{+DWL} and $S_{control}$.



Figure A1. Layer thicknesses for S_{+DWL} and $S_{control}$ for the upper 150 m.



Figure A2 shows vertical temperature profiles for a GOTM run with 0.1 m uniform vertical resolution.



Figure A2. (a) As in Figure 1, but with uniform 0.1 m thick vertical layers. (b) Difference between the temperature profiles in Figure 1 and (a).

Appendix B: Average Δ SST as Proxy for DSA

Figure B1 shows the field of Δ SST on the detection day.



Figure B1. Δ SST on detection day.

Data Availability Statement

Detailed information concerning the ICON model is contained in the publication Hohenegger et al. (2023). The ocean model GOTM is documented in Umlauf et al. (2005) and can be installed from https://gotm.net. The ERA5 data set used in this study has been provided by the Climate Data Store. The versions of ICON and GOTM used to produce the output, the run scripts, as well as the code that produces the figures can be found in https://hdl.handle.net/21.11116/0000-000C-1447-E.

References

- Acquistapace, C., Meroni, A. N., Labbri, G., Lange, D., Späth, F., Abbas, S., & Bellenger, H. (2022). Fast atmospheric response to a cold oceanic mesoscale patch in the north-western tropical Atlantic. *Journal of Geophysical Research: Atmospheres*, 127(21), e2022JD036799. https://doi. org/10.1029/2022JD036799
- Bellenger, H., & Duvel, J.-P. (2009). An analysis of tropical ocean diurnal warm layers. *Journal of Climate*, 22(13), 3629–3646. https://doi. org/10.1175/2008JCL12598.1
- Brilouet, P.-E., Redelsperger, J.-L., Bouin, M.-N., Couvreux, F., & Lebeaupin Brossier, C. (2021). A case-study of the coupled ocean-atmosphere response to an oceanic diurnal warm layer. *Quarterly Journal of the Royal Meteorological Society*, 147(736), 2008–2032. https://doi.org/10.1002/qj.4007

Acknowledgments

This work used resources of the Deutsches Klimarechenzentrum (DKRZ) granted by its Scientific Steering Committee (WLA) under project ID bm1253. This paper is a contribution to the project L4 ("Energy-Consistent Ocean-Atmosphere Coupling") of the collaborative research centre TRR 181 "Energy Transfers in Atmosphere and Ocean" funded by the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation)—Project-ID 274762653. Open Access funding enabled and organized by Projekt DEAL.



- de Szoeke, S. P., Marke, T., & Brewer, W. A. (2021). Diurnal Ocean surface warming drives convective turbulence and clouds in the atmosphere. Geophysical Research Letters, 48(4), e2020GL091299. https://doi.org/10.1029/2020GL091299
- Fu, R., Genio, A. D. D., & Rossow, W. B. (1990). Behavior of deep convective clouds in the tropical Pacific deduced from ISCCP radiances. *Journal of Climate*, 3(10), 1129–1152. https://doi.org/10.1175/1520-0442(1990)003<1129:bodcci>2.0.co;2
- Gentemann, C. L., Donlon, C. J., Stuart-Menteth, A., & Wentz, F. J. (2003). Diurnal signals in satellite sea surface temperature measurements. Geophysical Research Letters, 30(3), 1140. https://doi.org/10.1029/2002GL016291
- Hohenegger, C., Korn, P., Linardakis, L., Redler, R., Schnur, R., Adamidis, P., et al. (2023). ICON-Sapphire: Simulating the components of the Earth system and their interactions at kilometer and subkilometer scales. *Geoscientific Model Development*, 16(2), 779–811. https://doi. org/10.5194/gmd-16-779-2023
- Hohenegger, C., Kornblueh, L., Klocke, D., Becker, T., Cioni, G., Engels, J. F., et al. (2020). Climate statistics in global simulations of the atmosphere, from 80 to 2.5 km grid spacing. *Journal of the Meteorological Society of Japan. Ser. II*, 98(1), 73–91. https://doi.org/10.2151/ jmsj.2020-005
- Kawai, Y., & Wada, A. (2007). Diurnal sea surface temperature variation and its impact on the atmosphere and ocean: A review. Journal of Oceanography, 63(5), 721–744. https://doi.org/10.1007/s10872-007-0063-0
- Matthews, A. J., Baranowski, D. B., Heywood, K. J., Flatau, P. J., & Schmidtko, S. (2014). The surface diurnal warm layer in the Indian Ocean during Cindy/dynamo. Journal of Climate, 27(24), 9101–9122. https://doi.org/10.1175/JCLI-D-14-00222.1
- Price, J. F., Weller, R. A., & Pinkel, R. (1986). Diurnal cycling: Observations and models of the upper ocean response to diurnal heating, cooling, and wind mixing. *Journal of Geophysical Research*, 91(C7), 8411–8427. https://doi.org/10.1029/JC091iC07p08411
- Siebesma, A. P., Bretherton, C. S., Brown, A., Chlond, A., Cuxart, J., Duynkerke, P. G., et al. (2003). A large eddy simulation intercomparison study of shallow cumulus convection. *Journal of the Atmospheric Sciences*, 60(10), 1201–1219. https://doi.org/10.1175/1520-0469(2003)60 <1201:alesis>2.0.co:2
- Soloviev, A., & Lukas, R. (2013). The near-surface layer of the ocean: Structure, dynamics and applications. Springer Netherlands.
- Stevens, B., Bony, S., Farrell, D., Ament, F., Blyth, A., Fairall, C., et al. (2021). EUREC⁴A. Earth System Science Data, 13(8), 4067–4119. https:// doi.org/10.5194/essd-13-4067-2021

Sverdrup, H., Johnson, M., & Fleming, R. (1942). The oceans, their physics, chemistry, and general biology. Prentice-Hall.

- Umlauf, L., Bolding, K., & Burchard, H. (2005). GOTM—Scientific documentation. Version 3.2 (Vol. 63). Leibniz-Institute for Baltic Sea Research.
- Vial, J., Vogel, R., Bony, S., Stevens, B., Winker, D. M., Cai, X., et al. (2019). A new look at the daily cycle of trade wind cumuli. Journal of Advances in Modeling Earth Systems, 11(10), 3148–3166. https://doi.org/10.1029/2019MS001746
- Voldoire, A., Roehrig, R., Giordani, H., Waldman, R., Zhang, Y., Xie, S., & Bouin, M.-N. (2022). Assessment of the sea surface temperature diurnal cycle in CNRM-CM6-1 based on its 1D coupled configuration. *Geoscientific Model Development*, 15(8), 3347–3370. https://doi. org/10.5194/gmd-15-3347-2022
- Wick, G. A., & Castro, S. L. (2020). Assessment of extreme diurnal warming in operational geosynchronous satellite sea surface temperature products. *Remote Sensing*, 12(22), 3771. https://doi.org/10.3390/rs12223771